

RESEARCH PROTOCOL

Children with OCD:

Identifying Acceptable Support Strategies for Parents

(CO-ASSIST)



The University of Manchester





Children with OCD: Identifying Acceptable Support Strategies for Parents (CO-ASSIST)

1. PLAIN ENGLISH SUMMARY	.4
2. BACKGROUND AND RATIONALE	.4
3 AIMS AND OBJECTIVES	.5
3.1 Aim To undertake development work to better support parents and carers of children and young people with OCD.	5
3.2 Objectives	.5
3.3 Outcomes	.5
6 STUDY DESIGN	.6
6.1 Methods	.6
6.1.1 PHASE 1	.6
Systematic review	.6
Qualitative interviews with parents/carers (and optional journal)	.6
Focus Groups (or qualitative interviews) with professionals	.6
Service mapping	.7
6.1.2 PHASE 2	.7
6.1.3 PHASE 3	7
6.2 Data Collection	7
6.2.1 PHASE 1	8
6.2.2 PHASE 2 & 3	8
6.2.3 Demographic data	9
6.3 Data Analysis	9
6.3.1 PHASE 1	.9
6.3.2 PHASE 2	9
6.3.3 PHASE 3	10
7 STUDY SETTING	10
8 SAMPLING AND RECRUITMENT	10
8.1 Inclusion criteria	10
8.2 Exclusion criteria	10
8.3 Sampling	11
8.3.1 Size of sample	11
8.3.2 Sampling technique	11
8.4 Recruitment	12





8.4.1 Sample identification128.4.2 Consent129 ETHICAL AND REGULATORY CONSIDERATIONS139.1 Assessment and management of risk149.3 Peer review149.4 Patient & Public Involvement149.6 Data protection and patient confidentiality1510 DISSEMINATION POLICY1611 Project Timetable1612. References1711 APPENDICES18





Study Title	Children with OCD: Identifying Acceptable Support Strategies for Parents (CO-ASSIST)					
Short Title	Children with OCD: Identifying Support Strategies for Parents					
Study Design	A mixed-method qualitative study involving three phases: Phase 1 will include -in-depth semi-structured remote telephone or videoconference interviews (and an optional journal) with parents/carers - videoconference focus groups (or individual telephone or videoconference interview) with professionals. Phase 2- 21/2 hour remote synthesis meeting (attended by the CO-ASSIST team & a 2 1/2 hour remote consensus workshop event attended by mixed stakeholdersPhase 3 2-hour remote workshop will attended by mixed stakeholder events. These events are aimed at i) synthesising new and existing evidence to derive a roadmap of support resources and strategies, ii) identifying components of a future intervention and develop a protocol and outcome measures for its future development and testing.					
Study Participants	-Parents/carers of children and young people aged 8-18 with OCDProfessionals involved in supporting children and families with OCD -Charity representatives or OCD researchers involved in service development for families with OCD					
Planned Sample Size	 Phase 1 25-30 patients/carers (interviews) 3-4 focus groups (of 6-8 people) with professionals who have experience of working with families with a child with a diagnosis of OCD (may include some individual interviews according to professional availability/preference) Phase 2 18-25 key stakeholders (including parents/carers, professionals, charity representatives and OCD researchers) Phase 3 10 key stakeholders (including parents/carers, professionals, charity representatives and OCD researchers) Total estimated aggregate sample across 3 phases N=95 					
Follow-up Duration	No follow up required					
Planned Study Period	20 months: 01/04/2020 to 31/11/2021 (including 3 month extension to account for delays due to the pandemic)					
Objectives	 Primary aim To undertake development work to better support parents and carers of children and young people with OCD. Objectives To systematically review the evidence on interventions to support parents/carers of children with emotional disorders (<i>HRA approval not required</i>) To determine parent and professional (including professionals outside of the NHS) experiences, preferences and perspectives on support needs (<i>HRA approval required</i>) To undertake a mapping of local service provision (<i>HRA approval not required</i>) To identify key elements of potential support strategies and resources that are feasible and meet parental/carer needs (<i>HRA approval required</i>) 					
Funder	National Institute for Health Research (NIHR) Research for Patient Benefit Programme. NIHR ref no: 200720					
Sponsor	University of Manchester FBMH - Faculty research governance Team Carys Bannister Building Tel: 0161 2755436 email: fbmhethics@manchester.ac.uk					
Lead R&D	Sue Dobson: R& I Grants Co-ordinator Greater Manchester Mental Health Foundation Trust Tel: O161 276 3310, email: susan.dobson@mhsc.nhs.uk					





The University of Manchester

Chief Investigator	Dr Rebecca Pedley University of Manchester, Division of Nursing Midwifery Social Work (Jean McFarlane Building, Oxford Road Tel: 0161 306 7331 email: Rebecca.Pedley@manchester.ac.uk						
Co-Investigator	Professor Lovell : University of Manchester, Division of Nursing Midwifery Social Work Jean McFarlane Building, Oxford Road Tel: 07850207737 email:karina.lovell@manchester.ac.uk						
Co-Investigator	Professor Penny Bee University of Manchester, Division of Nursing Midwifery Social Work, Jean McFarlane Building, Oxford Road email: Penny.E.Bee@manchester.ac.uk Tel: 0161 3067652						
Research Associate	Dr Emma Sowden University of Manchester Division of Nursing Midwifery Social Work Jean McFarlane Building, Oxford Road Tel:07795 612178 email: emma.sowden-2 @manchester.ac.uk						

1. PLAIN ENGLISH SUMMARY

Obsessive-compulsive disorder (OCD) frequently begins in children and young people. Parents/carers have to deal with the responsibility of helping their child to access treatment and manage their mental health problem. Parents/carers often find their support role difficult, leading them to become distressed and overwhelmed. Many parents/carers find that their lives become controlled by their child's symptoms, as they become drawn into helping with repetitive behaviours and rituals.

Consultation with UK OCD charities and parent/carers tell us that despite parents'/carers' important role, they are often given little or no support. There is currently little evidence on how parents/carers and professionals view parental/carer support needs, or the help they would find most useful.

This study will use qualitative methods to understand what support is needed, or the help that would be most useful and feasible. These findings will be combined with existing evidence and mixed stakeholder synthesis to identify workable ways of providing support that will meet the needs of parents/carers of young people with OCD.

2. BACKGROUND AND RATIONALE

Relatives of people with OCD report experiencing significant levels of burden (1) shown by some studies to be comparable to that experienced in families of people with psychosis (2). Obsessive-compulsive disorder (OCD) frequently begins in children and young people (3); therefore, its impact particularly affects the parents and carers of children and young people (CYP) with OCD. Parents'/carers' role as the primary provider of practical and emotional support to their children means that these individuals are likely to experience these burdens differently to other family member groups (4). Parents/carers have to deal with the responsibility of helping their child to access treatment and manage their mental health problem. This role is often difficult, leading to parents feeling distressed and overwhelmed. Many parents/carers find that their lives become controlled by their child's symptoms, as they become drawn into helping with repetitive behaviours and rituals (5). Access to NICE recommended treatments for OCD, remains a challenge for families living in the UK due to long waiting lists for Child and Adolescent Mental Health Services (CAMHS). For parents/carers of young people





experiencing OCD, these factors may ultimately mean that their caring role is challenging, long-term and disproportionate to their child's age (4).

This is of significant concern given parents'/carers' vital role in supporting their child with their mental health problem. Furthermore, parents'/carers' own health outcomes and that of their wider family are likely to be negatively affected by the strain of living with high levels of burden induced by OCD (6). Parents of CYP with OCD have been found to experience significantly higher levels of mental health symptoms compared with parents who have a child without mental health problems (7). These findings emphasise the need to support parents/carers, both to prevent the development of mental health problems and to prevent the deterioration of mental health problems in those who are already unwell. Supporting parents/carers to cope more adaptively with the demands of supporting a child with OCD may reduce their levels of distress (8). In the long term, this is likely to bring economic benefits through reducing the chances of parents/carers becoming users of the NHS themselves and by increasing their likelihood of continuing in employment. Remaining in work is a significant challenge for mothers of children with OCD in particular, resulting in them feeling socially isolated by their caring role (9).

While the understanding of the detrimental effect that caring for a family member with OCD has on the family has grown in recent decades (5, 10) there is currently little evidence on how parents/carers and professionals view parental/carer support needs, or the help they would find most useful. There is a critical need to ensure parents/carers receive adequate and timely support when caring for a child with OCD to ensure their own well-being and to improve outcomes for children growing up with the mental health problem (6).

To deliver effective and acceptable support, it is first necessary to develop a detailed understanding of parents'/carers' needs and the forms of support that they would find helpful and acceptable. To our knowledge, no qualitative study has sought to identify the support needs and preferences of parents/carers of children with OCD. A qualitative methodology will enable an in-depth exploration of this previously neglected area. This mixed-methods study aims to use qualitative methods, existing evidence and mixed stakeholder synthesis to identify acceptable and feasible strategies and resources that could ultimately improve parent/carer outcomes through rigorous research, development and implementation.

3 AIMS AND OBJECTIVES

3.1 Aim

To undertake development work to better support parents and carers of children and young people with OCD.

3.2 Objectives

- To systematically review the evidence on interventions to support parents/carers of children with emotional disorders.
- To determine parent and professional (including professionals outside of the NHS) experiences, preferences and perspectives on support needs.
- To undertake a mapping of local service provision.
- To identify key elements of potential support strategies and resources that are feasible and meet parental/carer needs.

3.3 Outcomes





- Roadmap of stakeholder identified strategies and resources scored according to key intervention design criteria(11)
- Evidence and stakeholder informed identification of key components of a future intervention and an outline for its testing in a subsequent study.
- Outputs that can enhance parent/carer, academic, professional and lay understanding of OCD.

6 STUDY DESIGN

The proposed work will comprise a mixed-methods study conducted in three phases.

6.1 Methods

6.1.1 PHASE 1

Systematic review

We will conduct a systematic review of interventions to support the psychological well-being, coping, and quality of life of parents and carers of children and young people with emotional disorders. This review will ensure that Phase 2 and 3 are informed by existing evidence.

Qualitative interviews with parents/carers (and optional journal)

The Research Associate (RA) will conduct and audio-record remote (telephone or videoconferencing) interviews with 25-30 parents/carers of children with OCD. Interviews will be guided by a) a semi-structured interview schedule and b) use of a journal as a prompt (where available) to explore the challenges faced and to reflect on what might have helped. Parents will be invited to complete an optional journal of the challenges and needs they may have experienced in a week leading up to the interview. The journal will provide further depth and richness to the qualitative interview data. A topic guide has been developed in conjunction with a parent/carer co-researcher (who has a child with OCD) and consultation with members of OCD UK to ensure it is acceptable and comprehensive. The guide has also been informed by our previous systematic review (12, 13) and the Theoretical Framework of Acceptability (13). The interview will:

- Explore experiences and perceptions of current services in terms of their provision of support for parents/carers supporting a child with OCD.
- Identify parents'/carers' support needs, barriers, and enablers to accessing support.
- Explore parents'/carers' preferences and priorities for the content, delivery, and format of interventions to support parents in their caring role.

Focus Groups (or qualitative interviews) with professionals

The RA and the parent co-researcher will facilitate 3-4 remote focus groups (of 6-8 people) with professionals (via approved videoconferencing software) who have experience of working with families with a child with a diagnosis of OCD. Focus groups will provide an efficient yet powerful tool to explore topics that have had little previous investigation and provide a group dialogue to determine both individual and professional community influences on implementation. Professional focus groups will commence three months after parent interviews, to enable parent/carer interviews to inform focus group questioning. The topic guide will be informed by the needs emerging from parent/carer interviews, our past (12) and current systematic review and the Theoretical Framework of Acceptability (13). The topic guide will be prepared in conjunction with our parent/carer corresearcher and reviewed by a clinician within our team who is experienced in supporting families affected by OCD. The guide will elicit health professional perspectives on:





- Perceptions of current services offered to families with a child who has OCD.
- The needs of parents/carers who support a child with OCD
- Organisational and individual level barriers and enablers to providing parental/carer support
- The content, delivery, and format of support that would be acceptable to parents/carers and professionals.

Additional one-to-one telephone/video will also be offered to professionals who cannot, or do not want to, attend a focus group.

Service mapping

During Phase 1, alongside interviews and focus groups, we will contact all recruitment sites in which we are collecting data to ask about the services they currently offer to families supporting a child or young person with OCD. HRA approval is not required for this aspect of the study as we will not seek any data on the individuals using these healthcare services - we are only interested in the provision currently offered (e.g. whether parents are offered any education about OCD). We will also supplement this information with data obtained during professional focus groups or through background questions obtained during Phase 1.

6.1.2 PHASE 2

Parents/carers, health professionals experienced in treating CYP with OCD, OCD researchers, professionals from education settings (e.g., pastoral leads), and charity representatives will take part in a 2 ½ hour remote stakeholder consensus workshop. The sample will comprise a mixed group of 18-25 stakeholders (with roughly equal proportions of parents/carers to professionals) to ensure an optimal meaningful discussion. Prior to the stakeholder consensus workshop the CO-ASSIST research team (including co-investigators, parent co-researcher and charity representatives will attend a 2 ½ hour event aimed at synthesising learning from phase 1 and existing evidence from our systematic review to identify potential strategies and resources to enhance parental/carer support in CYP OCD. These strategies and resources (identified during the synthesis event) will be sense-checked at the stakeholder workshop through discussion and voting to score and reach consensus on key intervention design criteria for each strategy/resource generated.

Our methodology for synthesis in Phases 2 and 3 of the study has been guided by PPI consultation, experience in synthesis methodology (14) and experts in intervention development. Phase 2 is aimed at synthesising existing and newly emerging evidence (from Phase 1) to identify the aspects of support that key groups (parents/carers, professionals and charities) want to include in a new support package.

6.1.3 PHASE 3

Phase 3 of the study, will comprise a 2-hour meeting with the research team and a mixed group of key stakeholders (n=10), including parents/carers, professionals, charity representatives, and OCD researchers. Through reviewing the outcomes of Phase 2, this panel will select potential support strategies and resources to be included in a complex intervention. These strategies will be located within a larger logic model designed to inform the future evaluation of our intervention.

The final design of Phase 2 and 3 will be informed by our findings from Phase 1 and feedback from piloting workshop events and feedback from PPI.

6.2 Data Collection





Following the recent introduction of additional research governance approvals due to COVID-19 the research team has taken the decision to use remote/digital data collection across all phases of the study (avoiding face-face methods) to mitigate risks relating to COVID-19. We have included the option of postal invites and parents may have the option of receiving a hard copy of the parent journal. No personal identifiable information will be contained within the contents of the journal. All paper-based data will be converted to an electronic copy as soon as it is received and the paper copies will be securely destroyed. We have also included the option of photographing and emailing paper journals to minimise the use of postal return methods.

The research team has successful experiences (pre- COVID 19) of remote data collection for similar project designs and feels there is currently no strong reason why the project cannot be conducted effectively via remote methods. Furthermore, we have sought advice from OCD-UK and the lead NHS Trust who have been supportive of a remote design for this project.

6.2.1 PHASE 1

Parent interviews (or individual professional interviews) may take place via phone or video conference according to parent preferences. Focus groups will be conducted via videoconferencing software. Audio recordings for Phase 1 will be made using a University of Manchester provided encrypted audio recorder, or via University approved software platforms Microsoft Teams or Zoom. Video conferencing software will only be used where required. Associated video files, which are automatically created during the recording process (within videoconferencing software) will be saved to a University encrypted laptop C drive and will be permanently deleted immediately after the interview has ended. Only audio-recordings will be transferred to a secure drive (once checked) and then deleted from the temporary storage as soon as possible. Interview data will be transcribed as soon as possible by a University-approved transcription company or an approved member of the research team. During the transcribing phase or as quickly as possible on receiving the transcript, any identifiable information will be replaced with non-identifiable generic terms or pseudonyms. All files containing data relating to each participant will be given a unique ID number to ensure confidentiality at all times (pseudonymisation). The key that links this ID number to participant's personal information will be stored separately, and only approved members of the research team will have access to this key.

6.2.2 PHASE 2 & 3

The remote workshops for Phase 2 and 3 will be implemented by a team of experts in both intervention design and remote workshops. The workshops will be supported by the infrastructure and expertise of The University of Manchester Medical Education Team, IT services, and our co-investigator Professor Karina Lovell (who has extensive experience in conducting remote clinical services and research). We will pilot remote workshops in advance and provide training for all facilitators. Parent participants who are not familiar with video conferencing or would like the opportunity to trial it before the event will be provided with the opportunity to participate in a training/familiarisation event. OCD UK and the parent co-researcher will contribute to piloting the remote event.

A remote design will enable stakeholders to review materials and submit their preferences and their reasoning through online methods, which will be supplemented with video conferencing events according to participants' preferences and availability at the time. Participants who are unable to login into the remote workshop on the day will be able to submit their votes electronically at a time that is convenient for them. Throughout the workshop events, there will be opportunities for parents/carers to hold some discussions separately from professionals (with the option of separate breakout spaces if required). The final design of





these phases will be informed by our findings from Phase 1 and feedback from piloting events and feedback from PPI.

6.2.3 Demographic data

All participants will be invited to answer some background questions (demographic details). It will be explained to the participant that this information will help inform understanding of the data and will be anonymised and grouped together with other parent/carer data and will only be used individually to provide context to individual quotes e.g. mother/father/carer. It will be made clear to the participant that they are free to decline any questions

A series of background questions have been designed for both parents and professional participants. For parent participants in Phase 1 this will be collected verbally, either at the beginning or end of the parent interview according to preferences. For all other participants, they will be invited to answer some background questions as part of the electronic consent process using secure Manchester University approved software. All participants invited to complete electronic consent and background questions will be sent an email explaining the process (Appendix 3).

6.3 Data Analysis

6.3.1 PHASE 1

Anonymised transcribed data will be imported into the qualitative Data Analysis Software NVivo-QSR 12 on secure University servers to facilitate the analytical process. The analysis will take place on the anonymised data set only, and members of the research team will undertake the analysis.

Framework Analysis (15) will be used to analyse the data, as its matrix-based format will facilitate the sharing of data as a team. Framework analysis will be used to inductively and deductively code data. Inductive and deductive coding will be conducted, using the Theoretical Framework of Acceptability (13) and key concepts identified through our previous and current systematic review on factors associated with relatives' burden (12) and current support strategies. Completed journals will also be incorporated into the data set alongside participant transcripts during the analysis process to provide a source of 'in the moment' data. Ideas and themes will be discussed with investigators, parent co-researcher and charity representatives and reviewed to ensure that all ideas are included and that findings are credible and confirmable. Gaining wider team perspectives in qualitative research has been identified as an important activity through which new interpretations and insights can be gleaned (16). Parents and professionals interviews will initially be analysed separately, but we will then synthesise the findings from these two stakeholder groups for similarities and differences related to acceptable support strategies. The results of the analysis will be used to inform the consensus work involving mixed stakeholder events which will take part over two separate events (Phase 2 and Phase 3).

6.3.2 PHASE 2

During the synthesis event attended by the CO-ASSIST team we will use a matrix to summarise Phase 1 data where the key findings are represented as columns and the different sources of data by rows. We envisage that multiple theories will inform our selection of strategies and resources. Strategies and resources identified during the synthesis will form the focus of the stakeholder consensus workshop. Resources and presentation materials for the consensus workshop will be reviewed and approved by sponsor and NHS REC/HRA prior to the event. We will set a priori criteria informed by the RAND/UCLA Appropriateness





Method (17) and APEASE (11) for group consensus and undertake anonymised electronic voting, recording if and where consensus is reached. By the end of phase 2, we will have generated a roadmap of potential professional and parent/carer-derived support strategies; each scored in terms of its importance, necessity, practicality and acceptability to its end users. We will keep any design work produced during the workshop to inform our intervention development. All workshop material, will be anonymised, and any potentially identifying details will be removed from any design work produced during the workshop.

6.3.3 PHASE 3

We will identify a theory of change underpinning the intervention and a logic model for ensuing research, detailing the inputs, processes and outputs needed for successful intervention delivery and evaluation. We will specify the details of the intervention using the Template for Intervention Description and Replication (TIDieR) checklist (18).

7 STUDY SETTING

The size of the population of parents with children and young people experiencing OCD and professionals involved in supporting these families is limited. To maximise the opportunity for eligible individuals to take part, we plan to use a variety of research sites, including NHS and third sector organisations.

Data will be collected remotely (via video conferencing or telephone interviews), electronically or via postal return.

8 SAMPLING AND RECRUITMENT

8.1 Inclusion criteria

Parents/carers of children or young people aged 8-18 who have a diagnosis of OCD. Parents of children who have recently (defined as within the last year) exceeded the age of 18 or who are not currently in their caregiver role, yet whose parenting/caring experience is still reflective of the needs of current parents/carers of CYP in this age range, will also be included

A parents'/carers' self-reported acknowledgement of their child's formal diagnosis of OCD is adequate to meet to the objective of this study, avoiding the need for more formal potentially intrusive methods involving a formal proxy diagnostic measures or access to medical notes. Interpreters will be available for participants who are unable to speak English.

Professionals who have experience of working with Children and Young People with OCD (including OCD focused research). The same inclusion criteria will apply to all phases of the study.

Parents taking part in Phase 2 or 3 will require access to a computer, tablet or smartphone to join Zoom or Microsoft Teams conferencing software which will be used to facilitate Phase 2 and 3 of the study. Both Zoom and Microsoft Teams are supported by the University of Manchester. We will support parents in accessing this software.

8.2 Exclusion criteria





Parents of children and young people with OCD or professional who support families with OCD, who live outside of the UK. There will be no other exclusion criteria, other than the parents not being able to give valid consent, but we will not include capacity assessments in the study design. The same exclusion criteria will apply to all phases of the study.

8.3 Sampling

Sampling will be purposive and include several recruitment pathways to ensure variability in the sample and maximum inclusivity.

8.3.1 Size of sample

For qualitative interview/focus group studies, sampling is ideally judged complete when theoretical saturation has been reached, i.e. when increasing the sample size no longer contributes to new evidence and the data collected can sufficiently address the research aim. In practice, it can be difficult to demonstrate this point and it cannot be precisely specified in advance as it depends on the variability in the sample and the properties of the data. On going analaysis will be used to determine theoretical saturation and inform our final sample size. However from previous experience we expect this to be achieved with a maximum of 25-30 interviews.

The total sample will comprise the aggregate total of parent and professional participants from each Phase 1, 2 and 3 of the overall study (N=70-95). Participants recruited to each phase will comprise different participants; however, there may be some overlap. Participants who express an interest in participating in more than one phase will be supported to do this.

- Phase 1: 25-30 parents and 25-30 professionals
- Phase 2: 18-25 key stakeholders
- Phase 3: 10 key stakeholders

8.3.2 Sampling technique

We aim to recruit different parents and professional participants for each phase from a variety of sites and recruitment pathways (including a variety of NHS, third-sector channels, and social media channels). This sampling technique will ensure a diverse range of views, to reduce research burden, and to mitigate against selection bias. However if a participant was keen to participate in multiple phases we would support this.

Professionals will be purposively sampled to include multidisciplinary health professionals and professionals from other sectors outside the NHS (e.g., pastoral leads in education settings, clinicians working in the private and third sector and charity representatives from OCD charities). This will ensure a holistic understanding of parental/carer needs and the approaches that might address them. Sampling of families will aim to provide a balance of those parenting/caring for a young person (12-18), and those parenting/caring for younger children (8-11) are included, as the needs of parents of teenagers markedly differ from those parenting younger children. Sampling will also aim to ensure diversity in terms of the gender of parents and children, as well as their ethnicity. As there may be variation among parents/carers needs, according to their level of engagement with services, the sample will include parents/carers who have received a diagnosis of OCD but are still on the waiting list awaiting treatment (e.g., after initial assessment); those currently receiving treatment, and those who are currently not receiving treatment.





8.4 Recruitment

Recruitment pathways for each phase will be comparable; however, each phase will have its own set of study documentation (invitation letter, consent to contact form- paper and electronic, participant information sheet, participant ID, and consent form/process) designed according to each phase and participant group.

Potential recruitment pathways will include:

- a) Relevant NHS sites and third sector organisations that have agreed to act as research sites will advertise the study through posters and online methods.
- b) The study will be promoted through a project-specific website and social media accounts set up by the study team (through Twitter and social media). We will also advertise on the University of Manchester website/mail lists.
- c) Research sites (i.e. a member of the clinical team or a member of the Trust's research delivery team acting under arrangement with the responsible care organisation) may introduce the study either in person (only if the family is attending the service for routine care), via telephone or letter (postal or email -including a consent to contact form). The consent to contact form has been designed both as a hard copy and as an electronic version to reduce postal returns. Research sites will be provided with all study documentation including invitation letters, participant information leaflets and SAE. All study documentation will also be accessible via the study website in pdf form to reduce paper methods.

Potential participants will be made aware of CO-ASSIST's independence from the research site, that participation is voluntary, and that they can withdraw from the study at any time without giving a reason and without their care or their child's care or legal rights being affected. The research sites will only be responsible for identifying and signposting potential participants to the CO-ASSIST research team or obtaining 'consent to contact' if they are interested in participating or finding out more about the study. The CO-ASSIST research team will be responsible for recruitment of participants, including obtaining informed consent. Participants will be offered as much time as is needed to consider participation. Interested parents will be contacted by the RA to explain the study, answer questions and go through the informed consent form.

We anticipate that our primary recruitment pathway will be through online advertisements by displaying or distributing study posters/flyers at research sites, which will include both NHS and third sector organisations. This method will require participants to actively 'opt in' by contacting the research team if they wish to take part or find out more about the study.

8.4.1 Sample identification

Study advertisements will include contact details so that interested individuals can 'opt-in' to the study by contacting the researcher. The direct care team may assist in the identification of potential participants by reviewing families' medical records. The members of the CO-ASSIST team will <u>not have</u> access to family records (for screening for potential participants) or any contact details prior to participants giving 'consent to contact'.

8.4.2 Consent

Informed consent must be obtained prior to a participant undergoing any activities that are specifically for the purposes of the study. Following careful consideration and consultation with senior management teams within





the University of Manchester and NHS Trusts, a combination of audio-recorded verbal consent and electronic consent has been selected, as the most effective methods to ensure both the ethical conduct and safeguarding during the pandemic.

All potential participants will be provided with a CO-ASSIST Information Sheet, which has been designed according to the different requirements of the participating parties, i.e. parents, professionals/providers and according to the different phases of the study. Phase 2 and 3 have the same Participant Information Sheet for both parents and professionals. In addition to providing the information sheet, the researcher will briefly summarise the study paying particular attention to the voluntary nature of participation; the purpose of the study; what it will involve; any risks or benefits, their rights to decline or withdraw at any point and what will happen to the information they give us.

All participants will be given a minimum of 24 hours to consider the CO-ASSIST relevant Participant Information Sheet, and whether they would like to take part in the study. The RA will ensure that they are completely satisfied that the person fully understands the study and has had the opportunity to ask questions and have them answered before they can be asked to provide informed consent. Where participants appear to struggle to understand any aspect of the study, the researcher will repeat the information, trying to make the information as clear as possible. Participants will be given further time to consider the study if they appear unsure of their involvement.

Parents participating in Phase 1 of the study will be required to provide audio-recorded verbal consent before taking part in a qualitative interview. The consent will be recorded using a University of Manchester provided encrypted audio recorder or approved University video conferencing software, and the sound file will be stored separately to the qualitative interview. The associated video files for both the verbal consent and the qualitative interview will be deleted as soon as possible after completion of the interview. The RA will use a Standard Operating Procedure (SOP) and an electronic Pro-forma during the process of collecting verbal consent. A reference copy (electronic or paper according to participant preferences) of the consent details for each phase will be made available to all participants.

Informed consent for professionals interested in taking part in any of the three Phases will be obtained online through a secure electronic consent form designed within a University of Manchester approved software tool . Informed consent for parents interested in participating in Phase 2 and 3 will also be obtained online in the same way. Each participant will be assigned a unique ID before the completion of the online consent form and guidelines will be emailed to all participants. When they visit the link the first box will say 'please enter your unique ID', otherwise access the form will not be permitted. The options of Yes and No will be listed next to each statement of the Consent Form so that if a participant does not wish to consent to any optional point, they can clearly state this. There is also a measure in place that if a participant does not consent to the non-optional points, then they would be taken to a page thanking them for their participation but confirming that they could not take part in the study.

9 ETHICAL AND REGULATORY CONSIDERATIONS

The potential for improving understanding of how best to support parents/carers is high, though we cannot claim there will be a direct benefit for participants in the present study. However, the study will provide the opportunity for individuals with parental responsibility of a child or young person with OCD to share their experiences and support needs and shape the development of future support interventions.





9.1 Assessment and management of risk

The main study burden is expected to be the use of the participant's time. As different participants will be invited to take part in each phase of the study, the burden incurred is expected to be kept to a minimum. The study across each phase has been developed in collaboration with five PPI representatives, a parent corresearcher and members of OCD UK to ensure the design is acceptable and comprehensible for parents. Our previous experiences of carrying out interviews with relatives of people with OCD tells us that many relatives appreciate the opportunity to talk about their experiences, as they often feel disregarded. In addition, all parent participants for each phase of the study will be compensated for their time with the payment of gift vouchers (phase 1, 3) or cash payments (Phase 2) (which has been calculated at a rate of £25 per hour).

It is possible that discussions of parent support needs in their role of caring for a child with OCD will bring some quite strong emotions to the surface. A distress protocol will be employed, where the participant appears to start showing signs of distress. To help manage the situation sensitively, the RA will signal to participants that they have control over what and how much they say, reminding participants that they can decline to answer any questions if they wish, take breaks or stop the interviews at any point.

9.2 Research Ethics Committee (REC) and other Regulatory review & reports

This study has been designed according to the UK Policy Framework for Health and Social Care (HRA, 2017). As the study requires access to parents/carers of children and young people with OCD who may be attending NHS sites before the before the start of the study, a favourable opinion will be sought from the UK Health Departments Research Ethics Service NHS REC and the HRA. Substantial amendments that require review by NHS REC will not be implemented until that review is in place and other mechanisms are in place to implement at sites. All correspondence with the REC will be retained, and the CI will notify REC of the end of the study. An annual progress report (APR) will be submitted to the REC within 30 days of the anniversary date on which the favourable opinion was given, and annually until the study is declared ended. If the study is ended prematurely, the Chief Investigator will notify the REC, including the reasons for the premature termination. Within one year after the end of the study, the Chief Investigator will submit a final report with the results, including any publications/abstracts, to the REC.

9.3 Peer review

In order to secure funding for this project, the application process involved 2 stages of scientific review conducted by the East Midlands Regional Assessment Committee of the NIHR Research for Patient Benefit Programme. The grant was also subject to an internal review which is a mandatory part of the University's submission process for RfPB.

9.4 Patient & Public Involvement

Consultation with the public (parents/carers) and people with lived experiences will occur at all stages of the research process. PPI has informed the proposed project from its conception. A focus on the support needs of parents/carers of children with OCD was derived from early consultation with OCD UK, a service user-led charity and 13 parents/carers of CYP with OCD from a support group. A key message was the high levels of distress experienced by parents/carers, including depression and anxiety, which they attributed to the strain of their caring role and the lack of support. Parents/carers wanted greater recognition of the impact of OCD on the family. They emphasised the need for strategies and resources that would address both their child's needs and the needs of the family who support them.





The design stage has also been informed by further consultation with five parents/carers of CYP with OCD who are members of an online national support group. Individuals with lived experience of OCD include a charity representative who will be consulted for the duration of the project. They will assist in recruitment, interpretation of data, attend programme meetings and synthesis events and advising on effective dissemination and.

A parent/carer (who has a child with OCD) will form part of the research team working as a co-researcher over the course of the project. Our team is experienced in working with service user-researchers and carer researchers and has the resources, skills and experience to ensure they will receive adequate support, mentorship and research skills training, including qualitative methods (19).

The parent co-researcher will assist an experienced researcher in conducting focus groups and undertake double coding of a sub-set of the qualitative dataset. The parent co-researcher will not be required to undertake any research activities alone or obtain informed consent and will always be supported by a member of the research team. To enhance the study design, we have worked with our parent/carer co-researcher and OCD UK to develop study materials, including the parent/carer journal crib sheet/template, interview schedules, participant information leaflets, and study advertisements. The parent/carer co-researcher charity members will also work with the team to develop dissemination materials, ensuring the parent/carer voice is effectively communicated through our summaries and podcast.

9.6 Data protection and patient confidentiality

All members of the research team and study site staff will be required to comply with the requirements of the Data Protection Act 2018 and General Data Protection Regulation (GDPR) with regards to the collection, storage, processing and disclosure of personal information and will uphold the regulations core principles. All personal data will be collected electronically and will be stored on password-protected secure University servers according to our Data Management Plan, which is subject to the University of Manchester Research Data Management Policy and GDPR.

All data will be transferred to a secure University server as soon as is practically possible, and all other sources of data will be safely destroyed. All data (except for the participants consent form) will be coded and depersonalised, where the participant's identifying information is replaced with a unique ID number (comprised of an unrelated sequence of characters). The identification key enabling pseudonymisation of the data will be stored in a password protected Excel spreadsheet, separately from all other data. Only the CI, Research Associate and co-investigators will have access to this key.

Data collected during the study may be looked at by individuals from The University of Manchester, NHS Trusts or regulatory authorities for monitoring and audit purposes. Otherwise, only approved members of the study team will have access to a participant's personal data. The University of Manchester Research Data Storage (RDS) will be used for longer-term secure storage. University secure drives are backed up regularly and can only be accessed by approved members of the research team.

Consent details (including audio recorded consent and Select Survey consent) will be archived as essential documents until relevant research data are destroyed. Audio-recordings will be kept until the data has been transcribed, checked for accuracy and the early stage of analysis has been completed, after which time audio-recordings will be safely destroyed. Items such as contact details of participants will be deleted as soon as they





are no longer needed. Contact details will be retained by the Chief Investigator until analysis is completed, to make it possible to send participants (who indicated in their consent form) a summary of the study findings. However, personal data will be securely destroyed following dissemination of the study findings. The study Chief Investigator -Dr Rebecca Pedley will act as the custodian for the data generated from this study.

10 DISSEMINATION POLICY

We will maximise the impacts of this study by producing outputs that:

- a) Inform future intervention and service development through providing an evidence base of the strategies, resources and pathways that parents/carers and professionals perceive to meet parental/care need. Additionally, our consultation with OCD UK indicates that this work may help to support and inform their future funding bids (parent/carer benefit impact within 1-5 years – dependent on the strategies identified).
- b) Directly lead to a subsequent funding application to test the intervention (parent/carer benefit within 2-5 years dependent on the strategies selected and agreed by stakeholders).
- c) Inform members of the public about the problems faced by parents/carers and their needs; enhancing their understanding of OCD; a concern highlighted within our previous studies [25]and parent/carer consultation (patient/family member benefit impact - immediate impact)

A-C will be achieved through:

- Publication of an open-access journal article, co-authored by our parent/carer co-researcher
- Production of an executive summary, providing the output of the synthesis day, summarising the
 identified strategies and resources that need development and testing, together with
 parent/carer/professional ratings of key intervention design criteria including acceptability and
 practicability. The executive summary will be published online, shared via social media and sent
 directly to key charities and services.
- Production of a lay summary and podcast, co-developed with our parent/carer co-researcher, outlining the problems faced by parents/carers supporting CYP with OCD, their unmet needs and identified support strategies. Wide sharing through social media and relevant websites will enhance public awareness and understanding of OCD. We will also send the lay summary to study participants and will ask related charities to disseminate these via their website and social media networks.
- Sharing our findings at our synthesis day (including online remote engagement), through the participation of parents/carers as well as those directly involved in delivering and influencing patient care and support, including clinicians and charities working nationally.

An impact and dissemination plan will be developed at the start of the project and will remain on the agenda for project meetings.

11 Project Timetable

This study will span 20 months (including a 3 month extension to compensate for delays due to the pandemic), commencing with PPI involvement and collaboration with potential research sites in order to finalise and optimise the research design. We are aiming to commence parent interviews as soon as possible after approvals are in place (ideally before the end of September 2020) and November/December 2020 for





professional focus groups (and interviews). Phase 1 will end in May 2021. Phase 2 and 3 will recruit in August – October 2021. The study will close to recruitment in October 2021. October and November 2021 will be spent disseminating a summary of the research findings and planning for the next application. The research team will write up the findings and publish these as the study progresses. The findings from each phase will inform the next phase of the study.

12. References

1. Cicek E, Cicek IE, Kayhan F, Uguz F, Kaya N. Quality of life, family burden and associated factors in relatives with obsessive-compulsive disorder. General hospital psychiatry. 2013;35(3):253-8.

2. Kalra H, Nischal A, Trivedi JK, Dalal PK, Sinha PK. Extent and determinants of burden of care in Indian families: a comparison between obsessive-compulsive disorder and schizophrenia. International Journal of Social Psychiatry. 2009;55(1):28-38.

3. Albert U, Manchia M, Tortorella A, Volpe U, Rosso G, Carpiniello B, et al. Admixture analysis of age at symptom onset and age at disorder onset in a large sample of patients with obsessive-compulsive disorder. Journal of Affective Disorder. 2015;187:188-96.

4. Stengler-Wenzke K, Trosbach J, Dietrich S, Angermeyer MC. Coping strategies used by the relatives of people with obsessive-compulsive disorder. Journal of Advanced Nursing. 2004;48(1):35-42.

5. Cooper M. Obsessive-compulsive disorder: effects on family members. American Journal of Orthopsychiatry. 1996;66(2):296-304.

6. Steketee G. Social Support and Treatment Outcome of Obsessive Compulsive Disorder at 9-Month Follow-up. Behavioural Psychotherapy. 1993;21:81-95.

7. Derisley J, Libby S, Clark S, Reynolds S. Mental health, coping and family-functioning in parents of young people with obsessive-compulsive disorder and with anxiety disorders. The British Journal of Clinical Psychology. 2005;44(3):439-44.

8. Lazarus RS, Folkman S. Stress, Appraisal, and Coping. New York: Springer Publishing Company; 1984.

9. Futh A, Simonds LM, Micali N. Obsessive-compulsive disorder in children and adolescents: Parental understanding, accommodation, coping and distress. Journal of Anxiety Disorders. 2012;26(5):624-32.

10. Black D, Gaffney G, Schlosser S, Gabel J. The Impact of Obsessive-Compulsive Disorder on the Family: Preliminary Findings. The Journal of Nervous & Mental Disease. 1998;186(7):440-42.

11. Michie S, Atkins L, West R. The behaviour change wheel. A guide to designing interventions 1st ed Great Britain: Silverback Publishing. 2014:1003-10.

12. Pedley R, Wearden A, Berry K, Price O, Barber M, Bee P. Subjective burden and distress in relatives of people with obsessive-compulsive disorder: a systematic review. in preparation.

13. Sekhon M, Cartwright M, Francis JJ. Acceptability of healthcare interventions: an overview of reviews and development of a theoretical framework. BMC Health Services Research. 2017;17:88.

14. Bradshaw, T., Wearden A, Marshall M, Warburton J, Husain N, Pedley R, et al. Developing a healthy living intervention for people with early psychosis using the Medical Research Council's guidelines on complex interventions: phase 1 of the HELPER - InterACT programme. International Journal of Nursing Studies. 2012;49(4):398-406.

15. Ritchie J, Lewis J, Nicholls CM, Ormston R. Qualitative research practice: A guide for social science students and researchers: sage; 2013.

16. Barbour RS. Checklists for improving rigour in qualitative research: a case of the tail wagging the dog? BMJ. 2001;322:1115.

17. Fitch K, Bernstein SJ, Aguilar MD, Burnand B, LaCalle JR. The RAND/UCLA appropriateness method user's manual. Rand Corp Santa Monica CA; 2001.

18. Hoffmann TC, Glasziou PP, Boutron I, Milne R, Perera R, Moher D, et al. Better reporting of interventions: template for intervention description and replication (TIDieR) checklist and guide. BMJ : British Medical Journal. 2014;348:g1687.

19. Bee P, Brooks H, Callaghan P, Lovell K. A research handbook for patient and public involvement researchers: Manchester University Press; 2018.





20. Allen RES, Wiles JL. A rose by any other name: participants choosing research pseudonyms. Qualitative Research in Psychology. 2016;13(2):149-65.

11 APPENDICES

Amendment No.	Protocol version no.	Date issued	Author(s) of changes	Details of changes made
Draft for sponsor review	V.01	06.07.2020		
Second proof read of draft	V.02	14.07.2020	ESowden	Minor typographical changes
Final draft for submission to ethics	V1.0	18.08.2020	ESowden	Amendments made during sponsor review

A list of details of all protocol amendments will be detailed here here whenever a new version of the protocol is produced. Protocol amendments will be submitted to the Sponsor for approval prior to submission to the REC.